Author’s response to reviews

Title: Delivering an Optimised Behavioural Intervention (OBI) to people with low back pain with high psychological risk; results and lessons learnt from a feasibility randomised controlled trial of Contextual Cognitive Behavioural Therapy (CCBT) vs. Physiotherapy

Authors:

Tamar Pincus (t.pincus@rhul.ac.uk)
Shamaila Anwar (s.t.anwar@leeds.ac.uk)
Lance McCracken (lance.mccracken@kcl.ac.uk)
Alison McGregor (a.mcgregor@imperial.ac.uk)
Liz Graham (e.h.graham@leeds.ac.uk)
Michelle Collins (m.collinson@leeds.ac.uk)
John McBeth (j.mcbeth@keele.ac.uk)
Paul Watson (pjw25@le.ac.uk)
Stephen Morley (s.j.morley@leeds.ac.uk)
Juliet Henderson (juliet.henderson@rhul.ac.uk)
Amanda Farrin (juliet.hena.j.farrin@leeds.ac.ukc.uk)

Version: 2
Date: 14 November 2014

Author’s response to reviews: see over
Dear Editor,

WE have now inserted page numbers and a declaration about ethical approval in the methods section. The information about informed consent was already in the methods section.

Please accept for reviewing our manuscript entitled ‘Delivering an Optimised Behavioural Intervention (OBI) to people with low back pain with high psychological risk; results and lessons learnt from a feasibility randomised controlled trial of Contextual Cognitive Behavioural Therapy (CCBT) vs. Physiotherapy’.

We believe that the findings and lessons learnt from this study will be of interest to the journal readers. WE have followed the instruction to authors, and hope that our manuscript is found fit for publication.

Yours sincerely on behalf of the OBI team

Professor Tamar Pincus